Author's response to reviews

**Title:** Seasonal effects of influenza on mortality in a subtropical city

**Authors:**

Lin Yang (lyang@hkusua.hku.hk)
Chit Ming Wong (hrmrwcm@hkucc.hku.hk)
King Pan Chan (kpchanaa@hkucc.hku.hk)
Patsy Yuen Kwan Chau (patsy@hkucc.hku.hk)
Chun Quan Ou (ouchunquan@hotmail.com)
Kwok Hung Chan (chan.kwok.hung@hkucc.hku.hk)
J.S. Malik Peiris (malik@hkucc.hku.hk)

**Version:** 5  **Date:** 11 August 2009

Author's response to reviews: see over
Dear Editor,

Re: Seasonal effects of influenza on mortality in a subtropical city
Manuscript # 4404430412243797

Thank you very much for your email dated 10 August, 2009, requesting us to proof-read our manuscript. We have now revised the format according to your suggestions listed in email. The format of manuscript is now revised as required and resubmitted as in the document ‘Seasonal Effects_second revision_proof.doc’.

I should be grateful for your consideration for publication of our manuscript in the BMC Infectious Diseases.

Your sincerely,

Chit Ming Wong
Corresponding Author